

JGR Earth Surface

RESEARCH ARTICLE

10.1029/2021JF006397

Key Points:

- Modeled barriers with shrubs tend to be narrower, migrate landward more slowly, and be more strongly skewed volumetrically toward the ocean
- Shrubs increase the likelihood of discontinuous barrier retreat and, under some conditions, increase barrier vulnerability to drowning
- Shrubs expand more slowly across barriers experiencing greater dune erosion and overwash disturbance (and vice versa)

Supporting Information:

Supporting Information may be found in the online version of this article.

Correspondence to:

I. R. B. Reeves, reevesi@live.unc.edu

Citation:

Reeves, I. R. B., Goldstein, E. B., Moore, L. J., & Zinnert, J. C. (2022). Exploring the impacts of shrub-overwash feedbacks in coastal barrier systems with an ecological-morphological model. *Journal of Geophysical Research: Earth Surface*, 127, e2021JF006397. https://doi. org/10.1029/2021JF006397

Received 13 AUG 2021 Accepted 17 FEB 2022

Author Contributions:

Conceptualization: Ian R. B. Reeves, Evan B. Goldstein, Laura J. Moore, Julie C Zinnert Data curation: Ian R. B. Reeves Formal analysis: Ian R B Reeves Funding acquisition: Laura J. Moore, Julie C. Zinnert Investigation: Ian R. B. Reeves Methodology: Ian R. B. Reeves, Evan B. Goldstein, Laura J. Moore, Julie C. Zinnert Software: Ian R. B. Reeves, Evan B. Goldstein Visualization: Ian R. B. Reeves Writing - original draft: Ian R. B. Reeves Writing - review & editing: Ian R. B. Reeves, Evan B. Goldstein, Laura J. Moore, Julie C. Zinnert

© 2022. American Geophysical Union. All Rights Reserved.

Exploring the Impacts of Shrub-Overwash Feedbacks in Coastal Barrier Systems With an Ecological-Morphological Model

Ian R. B. Reeves¹, Evan B. Goldstein², Laura J. Moore¹, and Julie C. Zinnert³

¹Department of Earth, Marine and Environmental Sciences, University of North Carolina at Chapel Hill, Chapel Hill, NC, USA, ²Department of Geography, Environment, and Sustainability, University of North Carolina at Greensboro, Greensboro, NC, USA, ³Department of Biology, Virginia Commonwealth University, Richmond, VA, USA

Abstract Shrubs are common – and presently expanding – across coastal barrier interiors (the land between the foredune system and back-barrier bay), and have the potential to influence barrier morphodynamics by obstructing cross-shore overwash flow. The ecological and geomorphological consequences of ecomorphodynamic couplings of the barrier interior, however, remain largely unexplored. In this contribution, we add an ecological module of shrub expansion and mortality to a spatially-explicit exploratory model of barrier evolution (Barrier3D) to explore the effects of shrub-barrier feedbacks. In our model simulations, we find that the presence of shrubs significantly alters barrier morphology and behavior. Over timescales of decades to centuries, barriers with shrubs (relative to those without) tend to be narrower, migrate landward more slowly, and have a greater proportion of subaerial volume distributed toward the ocean-side of the barrier. Shrubs also tend to increase the likelihood of discontinuous barrier retreat, a behavior in which a barrier oscillates between periods of transgression and relative immobility, because shrubs induce prolonged periods of barrier immobility by obstructing overwash flow. However, shrubs can increase barrier vulnerability to drowning by preventing periods of transgression needed to maintain barrier elevation relative to rising sea levels. Additionally, physical barrier processes influence shrub expansion in our simulations; we find that greater dune erosion and overwash disturbance tends to slow the rate of shrub expansion across the barrier interior. Complementing recent observational studies of barrier islands in Virginia, USA, our results suggest that interior ecology can be a key component of barrier evolution on annual to centurial timescales.

Plain Language Summary Shrubs are common on coastal barriers, typically found sheltered behind dunes. During large storms, sand from the beach is washed into the barrier interior and beyond in a process called overwash. Overwash tends to cause the barrier to gain elevation relative to sea level, and also move the whole barrier landward. Shrubs, however, can obstruct this flow across a barrier. Here we explore how shrubs blocking overwash can impact barrier change, and how barrier change affects the growth of shrubs. We do this by adding new code to represent shrubs to an existing computer model of barrier evolution. Results from the computer model suggest that barriers with shrubs are typically narrower and tend to move landward more slowly compared to barriers without shrubs. Additionally, barriers with shrubs are more likely to move landward in a stop-and-go manner, switching between periods of fast landward motion and no motion. Therefore, barriers with shrubs are less likely to smoothly and continuously move landward. We also find that shrubs expand more slowly across barriers when storms cause significant dune erosion and overwash disturbance. Our work suggests that including shrubs in barrier models helps us better understand the ecological and geomorphological change of barrier systems.

1. Introduction

Ecological and geomorphological processes in coastal barrier systems are tightly coupled (e.g., Durán & Moore, 2013; Godfrey, 1977; Godfrey et al., 1979; Goldstein et al., 2017; Hesp, 2002; Keijsers et al., 2016; Rastetter, 1991; Reeves et al., 2020; Roman & Nordstrom, 1988; Stallins & Parker, 2003; Walters et al., 2014; Zinnert et al., 2017, 2019), owing to the similar spatiotemporal scales over which they operate. Barrier islands and spits comprise 10% of the world's continental coastline (Stutz & Pilkey, 2011), support rich ecosystems, buffer the impacts of storms on coastal regions, and are culturally and economically valuable hosts of human habitation and infrastructure. The future of barrier systems, however, remains uncertain with projected accelerated



relative sea-level rise (RSLR) (Sweet et al., 2017) and changes in storminess, such as increased storm intensity (e.g., Knutson et al., 2020). This uncertainty is further complicated by ecological transformations resulting from global climate change, which have become increasingly apparent in barrier systems in recent decades (Goldstein et al., 2018; Lucas & Carter, 2010; Osland et al., 2016; Zinnert et al., 2016).

Over centurial to millennial timescales, barriers tend to maintain their elevation relative to rising sea level by migrating upward and landward. On shorter timescales, some barriers can also experience intermittent periods of progradation and aggradation (e.g., Ciarletta et al., 2019; van IJzendoorn et al., 2021). Landward barrier migration is facilitated primarily by the process of overwash, whereby sediment eroded from the shoreface and beach is transported landward of the dune crest during high-water events (e.g., Dolan & Godfrey, 1973; Donnelly et al., 2006). As overwash flows landward over the interior of a barrier, flow velocity slows from lateral dispersal, percolation, and frictional drag (e.g., Schwartz, 1982), causing sediment to deposit on and potentially behind the barrier and therefore allowing the barrier to aggrade and translate up slope. Foredune height plays a central role in regulating the volume of overwash flux into the barrier interior (Houser et al., 2018; Rogers et al., 2015; Sallenger, 2000), such that barriers tend to transgress when dunes are low but are relatively stationary when dunes are tall (e.g., Reeves et al., 2021). Over decades to centuries, repeated cycles of dune loss and regrowth can lead to discontinuous barrier retreat (Reeves et al., 2021), a behavior in which barriers oscillate between periods of relative immobility and landward migration. This is in contrast to continuous transgression, the behavior in which barriers consistently retreat over time. Barriers can drown if overwash fluxes are insufficient for maintaining barrier geometry or elevation relative to sea level (e.g., Lorenzo-Trueba & Ashton, 2014; Mellett & Plater, 2018; Moore et al., 2010; Passeri et al., 2020).

An understanding of how overwash processes operate within the interior of a barrier – the land between the foredune and back-barrier marsh or bay – is critical for understanding barrier response to climate change (Zinnert et al., 2016, 2017, 2019). Most observational and modeling studies of barrier evolution typically neglect or average-over interior processes and heterogeneities, despite the fact that the barrier interior often covers a large proportion of total barrier area and can be important in its ecogeomorphic connection to the sandy barrier front (Zinnert et al., 2019). In recent years, studies of interior barrier dynamics in coupled human-natural systems have gained traction (e.g., Lazarus et al., 2021; Magliocca et al., 2011; Rogers et al., 2015), but interior ecology as a control on barrier processes remains understudied. Passeri et al. (2018) add landcover classification to hindcast simulations of barrier response to hurricane events, and find that these additions improve the accuracy of their model in capturing observed morphologic change. However, this approach uses static landcover without dynamic, two-way feedbacks between the ecology and geomorphology, and operates over timescales insufficient for exploring long-term barrier behaviors.

Previous studies on bidirectional ecogeomorphic feedbacks have primarily focused on vegetated dunes (e.g., Biel et al., 2019; Hacker et al., 2019), marshes (e.g., Kirwan & Murray, 2007; Morris et al., 2002), and seagrass communities (e.g., Carr et al., 2018; Reeves et al., 2020), but barriers can also host late successional terrestrial communities such as shrublands and maritime forest that can potentially influence barrier morphodynamics. Shrubs in particular are likely to impact overwash processes because of their tall height, rigidity, and high biomass density near the sediment bed, and are therefore the focus of this work. Earlier studies have qualitatively recognized the ability of woody vegetation to limit overwash penetration during storms (Claudino-Sales et al., 2008; Morton & Sallenger, 2003; Wang & Horwitz, 2007; Williams, 2015; Zinnert et al., 2019). By slowing or preventing overwash flow across a barrier, shrubs may support barrier resistance to storms and perhaps build topographic relief (Zinnert et al., 2017). On the other hand, a reduction in sediment connectivity between the front (shoreface, beach) and back (interior, back-barrier bay) of a barrier in the presence of shrubs could theoretically limit its ability to aggrade and transgress in response to RSLR and future storms (i.e., reduce its resilience). The long-term (decadal to centurial) consequences of these interactions for barrier behavior and evolution have yet to be explored.

Understanding the spatial extent of shrubs on barriers is key to understanding how shrubs impact barrier morphodynamics (Zinnert et al., 2017). Shrub seed dispersal in barrier systems is primarily avian-based (Ehrenfeld, 1990; Shiflett & Young, 2010), and long-distance dispersal (i.e., >100 m from the invasion front) is common (Herrmann et al., 2016; Woods et al., 2019). Given the steep environmental gradients observed in barrier systems (Ehrenfeld, 1990; Enwright et al., 2019; Young et al., 2011) and the particular sensitivity of most shrubs to salinity, freshwater availability, and disturbance events, shrub expansion is strongly influenced by abiotic post-dispersal





Figure 1. Model domain and parameterization. (a) Map of Hog Island in the Virginia Coast Reserve (VCR), US Mid-Atlantic coast. (b) Example Barrier3D elevation grid, with model domains labeled; each black (red) star represents a single living (dead) shrub plant. MHW is mean high water. (c) Aerial imagery from 2016 of shrub cover on central Hog Island. (d) Dead shrubs on the ocean side of Cobb Island. (e) Schematic barrier cross-section illustrating important variables and metrics discussed in this paper.

processes and filtering (Woods et al., 2019). Foredune elevation is a key environmental filter for shrub expansion in barrier systems (Woods et al., 2019), as tall dunes offer protection from salt-spray and overwash (Miller et al., 2008). In the absence of adequate foredune protection, shrubs can also establish if sufficiently far from the ocean shoreline (Miller et al., 2008). Barrier topography is likewise an important control on shrub expansion, as shrubs are typically constrained to a narrow elevation range (Young et al., 2011) that balances the needs of proximity to the freshwater lens with limited exposure to salt-water flooding. While woody encroachment is often related to changes in subaerial barrier area and elevation (relative to sea level; Shao et al., 1998), such as those that occur via shoreline erosion or RSLR, shrubs can temporarily expand even on barriers that are losing subaerial land (Young et al., 2007; Zinnert et al., 2016).

Shrub cover worldwide has expanded across coastal (Battaglia et al., 2007; Huang et al., 2020; Lucas & Carter, 2010), terrestrial grassland (Van Auken, 2000; D'Odorico et al., 2012), desert (Peters et al., 2006), savannah (Stevens et al., 2017), and tundra ecosystems (Shaver et al., 2001; Tape et al., 2006) in recent decades. On the barrier islands of the Virginia Coast Reserve (VCR; Figure 1a), over 40% of land cover changed from grassland to woody shrub thicket between 1984 and 2011 (Zinnert et al., 2016). During this time period, shrub cover on Hog Island, a VCR barrier island, expanded parallel to the ocean shoreline at rates of up to 300 m yr⁻¹ (Woods et al., 2019). This ecological transformation is primarily a response to macroclimate warming and is enhanced by positive feedbacks related to shrubs modifying their own microclimate (D'Odorico et al., 2012; Huang et al., 2018; Wood et al., 2020). The accelerated expansion of shrubs in coastal systems in response to global warming suggests that shrubs could play an increasingly dominant role in overwash processes going forward.



The ability of shrubs to obstruct the landward flow of overwash suggests that interior ecology can play an important role in barrier evolution, while the existence of a threshold dune elevation for shrub growth suggests that shrub ecology is strongly dependent on barrier morphodynamics. The effects of these ecological-morphological couplings on barrier and shrub evolution remain unexplored over long (10^1-10^3 yr) timescales. Here we add an ecological module of shrub expansion and mortality to an existing spatially-explicit exploratory model of barrier evolution (Barrier3D; Reeves et al., 2021). Barrier3D tackles the scale separation between event-based and long-term models by explicitly yet efficiently simulating dune evolution, storm overwash, and a dynamically evolving shoreface in response to individual storm events and RSLR. As an important novelty of this work, the ecology and geomorphology in our model framework are bidirectionally coupled, dynamically evolving in response to each other and external forcing conditions. We use this coupled model first to examine how shrubs alter barrier morphology over time and impact the rate and style of barrier migration. Then, we explore the ways in which barrier dynamics influence the rates and patterns of shrub expansion under a range of anticipated future conditions. Rather than numerically predict the evolution of a particular location or setting, the goal of this work is to explore and explain the large-scale, decadal-centurial behaviors that arise from feedbacks between barrier ecology and geomorphology.

2. Model Development

Barrier3D (Reeves, 2021) is an exploratory model (Murray, 2003, 2013) that simulates the morphological evolution of a barrier segment over time scales of years to centuries, resolving spatially explicit cross-shore and alongshore topographic evolution. In the sections below, we provide a brief summary of Barrier3D followed by an introduction to the new shrub expansion and mortality module. Reeves et al. (2021) provide a complete description of the base model formulation.

2.1. Parameterization

By examining across wide ranges of key input values beyond what is observed in any one location, and given the simplistic nature of the model parameterizations, our experiments are designed to investigate coupled dynamics relevant to most barrier systems. To ground the model in reality and provide a common starting point for all simulations, we parameterize Barrier3D and the shrub module using data from Hog Island, located on the Eastern Shore of Virginia, USA (Figure 1a). Hog Island is part of the Virginia Coast Reserve (VCR), a Long-Term Ecological Research site owned by the Nature Conservancy. Since the mid-twentieth century, direct human impact on the barriers, marshes, and bays of the VCR has been minimal (Orth & McGlathery, 2012). Hog Island is a 12 km long, mixed-energy barrier island characterized by high relief relative to other VCR islands, with dune ridges typically 3-4 m above the NAVD 88 datum (Oster & Moore, 2009); consequently, it has been relatively stable and infrequently disturbed in recent decades (Wolner et al., 2013). The dominant shrub species in the VCR is Morella cerifera L. (Myricaceae), which can be found in coastal environments along the Gulf of Mexico and the US Atlantic from Florida to New Jersey. M. cerifera is an evergreen shrub that can grow to heights of 5-6 m in barrier environments and coalesce into monospecific thickets (Young et al., 1994). The shrub module is therefore parameterized specifically for M. cerifera, and we discuss the potential impacts of using different species in Section 5.1 below. The rate of RSLR in the VCR, calculated over the past four decades, is 5.5 mm yr⁻¹ (NOAA, 2021). Table 1 lists all variable and input parameter definitions, values, and sources referenced in the main text; for all other input parameter values used in our simulations from the Barrier3D base model, see Table S1 in Supporting Information S1.

2.2. Barrier3D

Barrier3D operates over a 10-by-10 m grid with a 1-year time step. While model parameters are updated once a year, a probabilistically determined number of storm events occur within each year. The barrier segment consists of one or more alongshore rows of dune cells at the front (ocean side) of the grid, backed by an interior domain with a predetermined constant alongshore length and dynamically changing cross-shore width (Figure 1b); dune and interior cells follow different sets of rules as described below. Due to complexities in modeling beach morphodynamics and to focus on dune and interior dynamics, Barrier3D does not explicitly model a beach,



Table 1

Model Parameters and Variables

Parameter/ Variable	Definition	Units	Value	Source						
New parameters introduced in shrub model										
ϕ	Shrub percent cover		Dynamic variable	N/A						
Λ	Proportion of overwash flow reduction through a shrub cell at full shrub cover		0.15	This work						
BurialLimit	Threshold proportion of shrub height that shrubs can be buried, beyond which shrubs are killed		0.75	Keller (2020)						
Disp_mu	Mean of lognormal distribution of seed dispersal distance		-0.721891	Woods et al. (2019)						
Disp_sigma	Standard deviation of lognormal distribution of seed dispersal distance		1.5	Woods et al. (2019)						
D _{shrub}	Threshold elevation of fronting dune needed for shrub establishment	m MHW	2.29	Woods et al. (2019)						
Female	Proportion of shrubs that are female		0.5	Hokkanen (2013)						
GermRate	Proportion of shrubs that successfully germinate		0.6	Young et al. (1994)						
H _s	Shrub height	m	Dynamic variable	N/A						
H _{s_max}	Maximum shrub height	m	5.3	Young et al. (1994)						
Q_{i_eff}	Effective discharge through cell impacted by shrub	${ m m^{3}\ hr^{-1}}$	Dynamic variable	N/A						
SalineLimit	Threshold discharge through a shrub cell beyond which immature (<1 year) shrubs are killed	$m^3 hr^{-1}$	5	Tolliver et al. (1997)						
Seedmin	Minimum seeds produced per year per shrub	count	1000	Kwit et al. (2004)						
Seedmax	Maximum seeds produced per year per shrub	count	10000	Kwit et al. (2004)						
SprayDist	Distance from ocean shoreline that shrubs can establish in absence of sufficient dune	m	170	Miller et al. (2008)						
T _{fruit}	Age at which female shrubs start producing seeds	years	5	Zinnert (pers. comm.)						
UprootLimit	Threshold depth that shrubs can be buried, beyond which shrubs are eroded	m	-0.2	Conn and Day (1993)						
Z _{shrub_min}	Minimum surface elevation for shrubs	m MHW	0.74	This work						
Z _{shrub_max}	Maximum surface elevation for shrubs	m MHW	1.84	This work						
Parameters from Barrier3D base model formulation referenced in main text										
MHW	Mean high water	m NAVD88	0.46							
Q _i	Overwash discharge received by landward neighbor cell i	${ m m^3}~{ m hr^{-1}}$	Dynamic variable							
<i>r</i>	Characteristic dune growth rate: average of r_{\min} and r_{\max}	yr^{-1}	0.30-0.90							
R _{high}	Highest elevation of the landward margin of runup (i.e., total water level)	m MHW	Dynamic variable							
R _{low}	Lowest elevation of the landward margin of runup	m MHW	Dynamic variable							
r _{max}	Maximum dune growth rate	yr^{-1}	0.55-1.15							
r _{min}	Minimum dune growth rate	yr^{-1}	0.05-0.65							
RSLR	Relative sea-level rise	mm yr ⁻¹	3–15							

instead assuming invariant beach width and slope at the ocean side of the dune cells (see Section 5.1 for a discussion on the limitations of this assumption).

Dune crest elevation is taken as the sum of the dune toe elevation, which remains fixed over time relative to sea level, and the height of the dune above its toe. Within each model time step, dune cells grow logistically toward a maximum dune height (Houser et al., 2015), with the shape of the logistic curve controlled by r, the intrinsic dune growth rate. Dune growth rates vary randomly alongshore from cell to cell between an r_{min} and r_{max} , with the characteristic dune growth rate (\bar{r}) for the barrier segment calculated as the mean of r_{min} and r_{max} . When varying r_{min} and r_{max} , we hold the range between the two values constant at 0.5 yr⁻¹ (Houser et al., 2015). RSLR



is treated using a Lagrangian frame of reference by reducing all elevations relative to a sea level fixed at zero. Because RSLR lowers dune elevations, dune cells effectively at their maximum dune height continue to grow vertically each year and maintain their elevation relative to sea level. Next, a probabilistically determined number of storms occur by random selection from a list of 10,000 synthetic storms, each described by three statistics: (a) Duration; (b) R_{high}, the highest elevation of the landward margin of runup (i.e., the total water level); and (c) R_{low}, the lowest runup elevation. Where water levels overtop the dune crest during storms, dune heights are reduced following an empirical predictor of dune erosion (Goldstein & Moore, 2016; Long et al., 2014) in which higher storm water levels tend to cause greater vertical dune erosion, and overwash flow is routed cell-by-cell over the interior of the barrier, carrying sediment with it (Murray & Paola, 1994, 1997). Finally, shoreline erosion or accretion results from a combination of: (a) RSLR; (b) the cumulative volume of sediment removed from the upper shoreface by overwash and dune growth; and (c) net sediment exchange between the upper and lower shoreface (Lorenzo-Trueba & Ashton, 2014). Shoreface sediment flux depends on the shoreface slope, which tends to dynamically adjust toward an equilibrium in response to perturbations (i.e., overwash and dune growth). When the ocean shoreline erodes landward one cell width, the front row of the dune field is removed and the first row of the barrier interior functionally becomes the back row of the active dune field.

For the experiments presented herein, the initial morphology of the barrier interior comes from a digital elevation model (DEM) of Hog Island (NOAA, 2017) discretized into 10-by-10 m cells. Dune height and dune toe elevations are based on values extracted from the Hog Island DEM, and wave hindcast data offshore Hog Island (USACE Wave Information Studies) and water elevations from the nearest tide gauge in Wachapreague, VA, are run through a copula-based multivariate sea storm model (Wahl et al., 2016) to produce the synthetic storms.

2.3. Shrub Module

In the new shrub module of Barrier 3D, shrub plants occupy cells within the barrier interior, grow in size with age, and disperse seeds, allowing shrubs to expand. Various physical processes and factors influence shrub expansion and can lead to mortality. Conversely, shrubs impact physical processes by reducing overwash flow through cells that are occupied by shrubs. In the following, we describe in detail the parameterizations governing shrub dynamics in the model.

2.3.1. Growth

Established shrubs grow at the beginning of each year, represented by annual increases in percent cover and height – until maximum values are reached. Following observations on Hog Island (Young, unpublished observations) shrub percent cover (ϕ) increases logarithmically, reaching full cover after 9 years of growth. Assuming an allometric relationship between shrub width and height, shrub height (H_s) is taken as a function of percent cover: $H_s = \phi H_{s_max}$, where H_{s_max} is the maximum shrub height when shrub cover is full. We set H_{s_max} to 5.3 m following the empirical observations of shrub heights on Hog Island from Young et al. (1994).

2.3.2. Seed Dispersal and Establishment

Shrubs expand across the interior domain via seed dispersal at the beginning of each modeled year. Shrubs must be female and greater than or equal to 5 years in age (J. Zinnert, personal communication, 2017) to produce seeds. Of the shrubs that meet these requirements, the fecundity (seeds yr^{-1}) of each plant is determined by random selection from a uniform distribution; we set the bounds for seed fecundity (1,000–10,000) following the empirical measurements of Kwit et al. (2004). Following Hokkanen (2013), each seed has equal probability of being female or male. The number of seeds dispersed is further constrained by a germination rate (60%; Young et al., 1994), which removes seeds that fail to germinate. For each successfully germinating seed, the drop location is determined by the dispersal distance, randomly sampled from a lognormal probability distribution (Woods et al., 2019), and the direction, chosen randomly between 1 and 360°.

Seed establishment is dependent on the local environmental conditions. A seed becomes a new shrub in its determined drop location if the receiving cell is (a) within the interior domain of the barrier; (b) unoccupied by a previously established plant; (c) fronted by a dune cell exceeding an empirical threshold height (approximately 2.75 m on Hog Island; Woods et al., 2019) or, alternatively, is greater than 170 m landward of the shoreline (Miller et al., 2008); and (d) within an empirical elevation range. Because Barrier3D does not model a beach, we divide the static dune toe elevation by the beach slope and round to the nearest decameter to use as a static beach width;



this beach width in our simulations is 40 m, and therefore shrubs can establish in the absence of sufficient dunes if greater than or equal to 130 m from the foredune crest location. To establish the empirical elevation range, we use remotely-sensed land-cover-classification polygons of shrubs on Hog Island from Zinnert et al. (2016) to extract a histogram of land surface elevations from the 2017 DEM, then take the 5th and 95th percentiles as our minimum and maximum elevations for shrubs, respectively.

2.3.3. Mortality

Shrubs in the model are subjected to disturbances and stressors related to a dynamically evolving physical environment, which can lead to mortality. Each cause of mortality in the model is described below. Once dead, a shrub remains temporarily in place and can continue to influence overwash flow routing (as explained in Section 2.3.4), but can no longer grow or disperse seeds. Shrub mortality via natural senescence is not explicitly included in the model, under the assumption that gaps in shrub cover that arise from natural senescence are rapidly recolonized (Crawford & Yaoung, 1998). Shrub mortality therefore relies solely on physical conditions and stressors described below.

2.3.3.1. Elevation Change

Shrub mortality occurs if the surface elevation, relative to mean high water (MHW), of a cell containing a living shrub falls outside the established elevation range (e.g., via RSLR). Additionally, if a dead shrub plant falls below MHW, the plant is removed entirely from the model domain.

2.3.3.2. Burial

To represent the effects of burial (e.g., via overwash deposition), a shrub is killed if deposition in the cell since the time of initial shrub establishment exceeds 75% of the plant's height (Keller, 2020); a shrub plant, dead or alive, is removed from the interior domain completely if buried past 100% of its height.

2.3.3.3. Uprooting

To represent the effects of uprooting (e.g., via overwash scouring), a shrub is removed from the barrier interior if erosion in the cell since the time of initial establishment exceeds a threshold depth. We set this threshold to 0.2 m, as a majority of *M. cerifera* root biomass on Hog Island is found in the top 0.2 m of soil (Conn & Day, 1993).

2.3.3.4. Saline Flooding

To account for the effects of short-duration saline flooding (i.e., overwash events) on immature shrubs (Tolliver et al., 1997), the model kills all one-year-old plants through which discharge greater than $5 \text{ m}^3 \text{ hr}^{-1}$ is routed.

2.3.3.5. Shoreline Erosion

During barrier migration, when the front row of the interior domain becomes the back row of the dune field, all shrub plants dead and alive are removed from the interior domain to represent the impacts of shoreline erosion.

2.3.4. Interaction With Overwash

Shrubs in the model obstruct the flow of overwash across the barrier. The discharge of water leaving a cell with a living shrub plant is reduced according to the maximum flow reduction coefficient (Λ) and the plant's percent cover:

$$Q_{i_eff} = Q_i(1-\Lambda)\phi$$

where Q_{i_eff} is the effective discharge leaving the cell in question for the neighboring cell *i*, and Q_i is the calculated discharge leaving for neighboring cell *i* in the absence of a shrub plant. In this way, the flow of water through a cell with a shrub plant is reduced (relative to if no shrub is present), with larger shrubs reducing more flow than smaller shrubs. In turn, a reduction in discharge leaving a cell tends to cause greater net deposition of sediment within the cell, as a greater proportion of sediment is transported into the cell than carried out. In the absence of empirical measurements for the maximum flow reduction coefficient, we test the sensitivity of this parameter by varying Λ from 0.05 to 0.35 (Figure S1 in Supporting Information S1), as discussed below in Section 3.



Table 2

Parameter Values for Model Experiments

Experiment		RSLR (mm yr ⁻¹)	$\bar{\boldsymbol{r}}$ (yr ⁻¹)	Duration (yrs)	Barrier segment length (m)	Initial dune heights (m)	Λ	Seed placement
Impacts of shrubs on barrier morphology and migration	Varying RSLR (Figures 2 and 3)	3, 6, 9, 12, 15	0.6	1,000	500	0.1–1.4	0.15	Entire barrier
	Varying \bar{r} (Figures 4 and 5)	9	0.3, 0.45, 0.6, 0.75, 0.9	1,000	500	0.1–1.4	0.15	Entire barrier
Geomorphological controls on shrub expansion	Varying RSLR (Figure 6)	3, 6, 9, 12, 15	0.6	150	5,000	1.0–1.4	0.15	First 100 m alongshore
	Varying \bar{r} (Figure 7)	9	0.3, 0.45, 0.6, 0.75, 0.9	150	5,000	1.0–1.4	0.15	First 100 m alongshore

In the model, dead shrub plants are only two-thirds as effective at blocking overwash flow as living shrubs. We derived this ratio by comparing "summer" versus "winter" Manning's n values from floodplains of "medium to dense brush" (Chow, 1959) as proxies for living and dead shrubs, respectively. Therefore, when water is routed through a cell with a dead shrub, Λ is multiplied by 0.66 and ϕ refers to the shrub's percent cover at time of death.

3. Impacts of Shrubs on Barrier Morphology and Migration

To assess the ways in which shrubs alter barrier morphology and the rate and style of barrier retreat, we run simulations with and without shrubs across broad ranges of two key input parameters: RSLR and characteristic dune growth rate (\bar{r}). Dune growth rate is an important factor in controlling shrub expansion and mortality and in determining the vulnerability of a barrier to storm impacts (Durán Vinent & Moore, 2015; Goldstein & Moore, 2016; Reeves et al., 2021). Varying dune growth rates is also akin to varying storm frequency – that is, faster growth rates reduce the time needed for a dune to ecover before the next storm, and vice versa. RSLR is fundamental to barrier evolution, and varies spatially while also depending on future carbon emissions (e.g., Sweet et al., 2017).

Each simulation runs for 1,000 years or until the barrier drowns, with the alongshore length of the barrier segment set to 500 m. A duration of 1.000 years is able to capture trends in barrier migration (Reeves et al., 2021) and equilibrium morphologies, and is also designed to control for the effects of stochastic processes happening on shorter time scales (not necessarily to quantitatively predict a millennium of morphological or ecological evolution). The initial dune height is set randomly for each simulation between 0.1 and 1.4 m (with 0 and 1.5 m as the minimum and maximum dune heights, respectively), with random white noise perturbations alongshore of ± 0.1 m. To allow shrubs to repopulate a barrier bereft of any fruiting plants, a single shrub seed is dropped randomly across the barrier interior each year. We use a relatively conservative value of 0.15 as the maximum flow reduction coefficient (Λ), which produces qualitatively realistic morphology. Varying Λ from 0.05 to 0.35 results in little difference in our results and does not qualitatively change our conclusions (Figure S1 in Supporting Information S1). See Table 2 for the values of parameters used in our experiments. We consider a barrier to have drowned when the maximum subaerial width of the interior domain thins to less than 10 m (one cell width), at which point the simulation ends. To determine the style of barrier retreat (i.e., continuous or discontinuous), we use an algorithm from Reeves et al. (2021): Discontinuous retreat occurs if the simulation includes two or more alternating periods of both transgression and relative immobility; otherwise, the behavior is considered continuous. Periods of immobility (transgression) are defined as 30 years or more of shoreline change rates under (over) 0.5 m yr⁻¹.

3.1. RSLR

In the first set of experiments we vary RSLR from 3 to 15 mm yr⁻¹ in increments of 3, while holding \bar{r} at 0.6 yr⁻¹. At this intermediate value of \bar{r} , dune heights tend to be bistable; that is, dunes tend to fluctuate between a highand low-elevation state (e.g., Durán Vinent & Moore, 2015) instead of remaining within one elevation state for the duration of the simulation. To account for storm stochasticity within each unique simulation, we run simulations at each RSLR rate 100 times and present the mean (n = 100) time series of different morphological metrics for each RSLR value (Figure 2).





Figure 2. Varying relative sea-level rise (RSLR). Averaged (n = 100 simulations) time series of key variables for barriers with shrubs (dashed lines) and without shrubs (solid lines), with lines colored by RSLR rate used in the simulations.

For each RSLR rate we explore, each of the average time series flatten toward a steady equilibrium value after approximately 200–500 years of adjustment (Figures 2a–2d and 2f). The trajectories of individual simulations do not reach any sort of equilibrium state, as a barrier in any one simulation is constantly changing in morphology and shrub cover as the simulation progresses through time. Instead, trajectories of individual simulations tend to oscillate around the equilibrium values of the averaged time series that they collectively produce (Figure S2 in Supporting Information S1). These steady states are not conditioned by the initial barrier morphology, for the average time series approach the same steady state values even if the simulations begin with much narrower initial barrier morphology (Figure S3 in Supporting Information S1). Across all RSLR rates, barrier widths, volumes, and elevations approach steady attracting values maintained consistently above zero both with and without shrubs, indicating that drowning is not a steady state under these particular forcing conditions. Only 1.0% of all





Figure 3. Discontinuous behavior with shrubs (green) and without shrubs (blue) across a range of relative sea-level rise rates. (a) Probability of discontinuous retreat, (b) average duration of immobile periods, and (c) average duration of transgressive periods. Error bars are one standard deviation.

simulations with shrubs, and 0.4% of simulations without shrubs, drown over the course of 1,000 years. However, there are significant morphological and behavioral differences between barriers with and without shrubs.

Overall, barriers with shrubs tend to be narrower, have less overwash flux, and migrate landward more slowly relative to barriers without shrubs (Figures 2a–2c). Additionally, barrier volume above MHW tends to be smaller with shrubs (Figure 2d), and distributed more toward the front (ocean-side) of the barrier (Figure 2e). With the effects of increased frictional drag, overwash through shrub thickets is deposited over shorter distances, resulting in volumetric distributions skewed more strongly toward the barrier front (Figure 2e). By building topography toward the front of the barrier, shrubs inhibit additional sediment transport into the barrier interior and therefore reduce volumetric overwash flux (Figure 2b). With reduced overwash fluxes deposited over shorter distances, barriers with shrubs tend to be narrower than barriers without (Figure 2a), and smaller average barrier volumes (Figure 2d) follow as a consequence of narrower widths. Lastly, because less sediment is removed from the shoreface/beach environment and deposited as washover in the interior, shrubs slow the landward translation of the ocean shoreline (Figure 2c). These findings are consistent across all RSLR rates.

Among barriers without shrubs, increasing the rate of RSLR tends to decrease the equilibrium barrier width (Figure 2a). This occurs because higher rates of RSLR increase landward translation of the ocean shoreline (e.g., Bruun, 1962) and seaward translation of the back-barrier shoreline, especially during periods of barrier immobility, and thereby narrow the barrier from both the front and back. The opposite outcome is observed for barriers with shrubs, which tend to be wider with increasing RSLR rates because overall shrub cover is lesser (Figure 2f). Higher RSLR rates reduce the available habitat for shrubs and lead to mortality via passive inundation and shoreline erosion. Thus, for the case of barriers with shrubs, the geomorphological effect of higher RSLR rates on shrub expansion and mortality (which tend to increase barrier width).

The presence of shrubs also increases the likelihood of discontinuous retreat and the average duration of immobile periods (Figure 3). Periods of immobility tend to be longer with shrubs (Figure 3b) because shrubs delay the onset of overwash flux after dunes transition to a low-elevation state. When dunes are tall, overwash is typically limited and the barrier tends to be immobile, which favors shrub expansion across the barrier interior. If dune heights are consequently reduced via vertical (storm) or lateral (shoreline change) erosion, shrubs limit overwash flux across the barrier that would otherwise lead to rapid barrier transgression. Since overwash flux is limited, the barrier tends to remain relatively immobile until shrub cover has been effectively removed by mortality and shoreline erosion; alternatively, shrubs may prevent immobile periods altogether if the dunes can recover to a high-elevation state before shrub cover is lost. This lag in the onset of transgression is longer at lower RSLR rates because shrub cover is greater; at higher RSLR rates, there are fewer shrubs to be removed before substantial overwash – and therefore transgression – can commence. Additionally, shrubs tend to reduce the average duration of transgressive periods (Figure 3c). This occurs because shrubs can establish in the absence of sufficient dune heights if the barrier is wide enough, eventually work their way to the front of the barrier via shoreline erosion (i.e., as the shoreline recedes toward them), and impede overwash, thereby potentially leading to immobility sooner. This process is less impactful at higher RSLR rates because barriers tend to be narrower, making it more difficult for shrubs to establish in the absence of dunes. Shrubs increase the probability of discontinuous retreat (Figure 3a) by providing the prolonged cessation of transgression needed to produce discontinuous behavior. This impact is greatest at higher RSLR rates which produce conditions that, in the absence of shrubs, rarely result in



barrier immobility (Reeves et al., 2021). The large variance in immobility and transgressive durations (Figures 3b and 3c) is a product of the stochastic nature of storms and their interactions with dune dynamics, emphasizing the importance of dune-storm interactions in controlling the evolution of barriers over decadal-centurial timescales (Reeves et al., 2021).

3.2. Characteristic Dune Growth Rates

Next, we vary \bar{r} (the characteristic dune growth rate) from 0.3 to 0.9 yr⁻¹ in increments of 0.15, while holding RSLR at 6 mm yr⁻¹, the present RSLR rate in the VCR and a conservative estimate of the near future global mean sea-level rise rate (Sweet et al., 2017). As before, we run 100 simulations at each \bar{r} and present the mean of these 100 simulations as our results.

At high dune growth rates ($\bar{r} = 0.75$ and 0.9 yr⁻¹), average barrier trajectories trend toward steady states of drowning (Figures 4a and 4c). For the highest rates of dune growth ($\bar{r} = 0.9$ yr⁻¹), 74% of simulations with shrubs drowned after an average of 562 years (± 200), while 38% of simulations without shrubs drowned after an average of 568 years (± 224). At low to intermediate RSLR rates ($\bar{r} = 0.3-0.6$ yr⁻¹), only 0.7% of barriers in all simulations drowned. Drowning in the model is more likely at higher dune growth rates because tall dunes limit washover deposition in the barrier interior and back-barrier bay, allowing for passive inundation of the barrier interior that outpaces the rate that overwash events (which occur infrequently when dunes are tall) are able to build elevation. Shrubs increase the vulnerability of barriers to drowning by further starving the barrier interior of washover deposition needed for the barrier to maintain its elevation relative to sea level.

Consistent with the RSLR simulations of Section 3.1, barriers with shrubs tend to be narrower, have less overwash flux, and migrate landward more slowly across all characteristic dune growth rates (Figures 4a–4c). Likewise, barrier volume with shrubs tends to be lesser and skewed more strongly toward the front (ocean-side) of the barrier (Figures 4d and 4e). Dune growth rate is a strong regulator of barrier width, with slower-growing dunes allowing greater cumulative overwash fluxes (Figure 4b) and therefore wider barriers (Figure 4a); smaller cumulative overwash fluxes at higher dune growth rates maintain narrower barriers. The difference in barrier width between barriers with shrubs and without tends to be greatest at intermediate dune growth rates, where shrub-overwash interaction is large enough to cause significant width differences, yet disturbances are small enough to allow significant shrub expansion across the barrier. These experiments also provide insight into the effects on barrier evolution on shrub ecology. For example, even though foredunes of a threshold height are necessary for the establishment of shrubs, the fastest dune growth rates lead to the least shrub cover (Figure 4f). This occurs because consistently tall foredunes result in barriers that are both narrower and lower in elevation, thereby limiting suitable shrub habitat. The greatest shrub cover over time is observed on barriers with intermediate dune growth rates, where the effects of overwash creating or maintaining habitat are balanced with the tendency of overwash to physically disturb the ecology.

At low to intermediate dune growth rates ($\bar{r} = 0.3-0.6 \text{ yr}^{-1}$), the presence of shrubs increases the likelihood of discontinuous retreat (Figure 5a). Periods of transgression are typically prevalent at low dune growth rates because dunes are slow to recover following disturbance; however, the presence of shrubs induces prolonged cessations of transgression needed to produce discontinuous behavior, thereby increasing the likelihood of discontinuous retreat. At high dune growth rates ($\bar{r} = 0.75-0.9 \text{ yr}^{-1}$), periods of immobility are common; therefore, the shrub lag in overwash flux either has little to no effect on the probability of discontinuous behavior, or can in fact reduce the probability of discontinuous behavior in favor of continuous immobility by preventing periods of transgression altogether.

At low to intermediate dune growth rates, shrubs also increase the average duration of immobile periods (Figure 5b) and reduce the average duration of transgressive periods (Figure 5c). Without shrubs, the average duration of immobile periods tends to increase with higher dune growth rates (Figure 5b, blue line) because taller dunes reduce overwash fluxes necessary for transgression. Interestingly, the opposite is observed with shrubs: The average duration of immobile periods tends to decrease with higher dune growth rates (Figure 5b, green line). This is because higher dune growth rates lead to narrow barrier interior widths and limited suitable shrub habitat, resulting in less shrub cover and less impact on immobile durations. Similarly, shrubs have a negligible effect on the average duration of transgressive periods at high dune growth rates (Figure 5c) because suitable habitat is





Figure 4. Varying characteristic dune growth rates (\bar{r}). Averaged (n = 100 simulations) time series of key variables for barriers with shrubs (dashed lines) and without shrubs (solid lines), with lines colored by \bar{r} used in the simulations.



Figure 5. Discontinuous behavior with shrubs (green) and without shrubs (blue) across a range of characteristic dune growth rates. (a) Probability of discontinuous retreat, (b) average duration of immobile periods, and (c) average duration of transgressive periods. Error bars are one standard deviation.





Figure 6. Rate and pattern of shrub expansion across a range of relative sea-level rise (RSLR) rates. (a and b): Average shrub front expansion rate as a function of (a) cumulative overwash flux and (b) average dune height, with points colored by the RSLR rate used in the simulation. (c–f): Box plots separated by RSLR rate of (c) average shrub front expansion rate, (d) number of years with no expansion of the shrub front, (e) average dune height, and (f) cumulative overwash flux.

limited. Regardless of these trends, immobile periods tend to be longer, and transgressive periods shorter, in the presence of shrubs.

4. Geomorphological Controls on Shrub Expansion

To explore the ways in which barrier morphologic evolution influences the rate and pattern of shrub expansion, we conduct two sets of experiments across the same ranges of RLSR (Figure 6) and characteristic dune growth rates (Figure 7) as in Section 3. Unlike the preceding simulations, however, the simulations in these experiments use a 5,000 m-long segment of Hog Island as the initial morphology and run for 150 years (Table 2), a duration that is long enough to capture a full cycle of shrub expansion and significantly reduces model run time given the ten-fold increase in barrier length relative to the preceding simulations. In addition, each simulation begins with





Figure 7. Rate and pattern of shrub expansion across a range of characteristic dune growth rates (\bar{r}). (a and b): Average shrub front expansion rate as a function of (a) cumulative overwash flux and (b) average dune height, with points colored by the value of \bar{r} used in the simulation. (c–f): Box plots separated by \bar{r} of (c) average shrub front expansion rate, (d) number of years with no expansion of the shrub front, (e) average dune height, and (f) cumulative overwash flux.

a foredune that is in a high-elevation state. The initial foredune height is chosen randomly between 1.0 and 1.4 m, and a single seed is dropped randomly each year on one end of the barrier within the first 100 m (2%) alongshore. Controlling for the location of the initial cluster of shrub establishment ensures that all simulations are able to expand the same maximum distance (here, 5 km). To account for storm and seed dispersal stochasticity, we run 100 simulations at each RSLR and \bar{r} value.

We calculate the alongshore location of the shrub front as the 95th quantile of the alongshore distribution of all cells with living shrubs, which helps capture the front of the main thicket while disregarding small outlying patches or singular plants. Additionally, we calculate an average shrub expansion rate (\bar{e}) as 90% of the barrier length (L_{end} ; i.e., the distance at which the change in shrub front location fully saturates) divided by the time it takes for the shrub front to reach L_{end} (t_{Lend}): $\bar{e} = L_{end}/t_{Lend}$. Because shrub expansion does not always begin





Figure 8. Examples of (a) rapid continuous and (b) slower interrupted expansion (at $\bar{r} = 0.3 \text{ yr}^{-1}$) of the shrub front across a range of characteristic dune growth rates (\bar{r}). All simulations were run with the same initial and forcing conditions; differences in the behavior of shrub expansion arise from the stochasticity in seed dispersal and storm disturbance dynamics.

immediately at the start of each simulation, t_{Lend} begins when the shrub front location exceeds the first 100 m of the barrier.

Slower shrub front expansion rates are correlated with dune erosion and overwash disturbance (Figures 6a, 6b, 7a and 7b). These disturbances slow the rate of shrub encroachment by both promoting environmental conditions within the barrier interior that restrict successful post-dispersal establishment (e.g., salinity intrusion), and by killing or eroding away shrubs (which also has the effect of reducing the number of seeds dispersed). If undisturbed, the expansion of the shrub front over time can be described as logarithmic, with slow initial expansion, followed by rapid acceleration, then deceleration as the shrub front approaches the end of the barrier segment (e.g., Figure 8a). Disturbance events can cause arrested expansion of the shrub front (e.g., Figure 8b), leading to prolonged periods with no net change, or sometimes negative change, in the alongshore location of the shrub front.

Overall, there is significant variance in shrub expansion rates across all parameter combinations. Despite such variance, shrub expansion tends to be

faster at lower RSLR rates (Figure 6c), and the number of years in which no positive shrub expansion is observed (i.e., no movement or negative movement of the shrub front location occurs) tends to be fewer (Figure 6d). This is because dunes are typically higher (Figure 6e) and overwash fluxes lower (Figure 6f) under these conditions. Higher rates of RSLR also lower a greater number of interior cells below the elevation threshold for shrub establishment, and therefore reduce expansion rates by limiting suitable habitat. At higher dune growth rates, shrub expansion tends to be faster (Figure 7c) and the number of years without positive expansion tends to be lower (Figure 7d). Higher dune growth rates reduce dune and overwash disturbances (Figures 7e and 7f) because dunes are more likely to recover before the next storm event.

5. Discussion

5.1. Model Limitations

The goal of this work is to investigate barrier and shrub dynamics and the ecological-geomorphological feedbacks that give rise to them, rather than to numerically predict or reproduce barrier/shrub evolution of a particular setting or under specific conditions. The parameterizations encoded in our shrub module are considerable simplifications of real-world processes, which may limit quantitative precision of model results. Yet, the assumptions and simplifications we have made are constrained by or derived from empirical data such that the compound effects of many processes operating at smaller spatiotemporal scales are represented. Basing the model on emergent variables (e.g., shrub elevation range) and interactions rather than the finer scale processes that collectively produce them is the most appropriate modeling approach for exploring and explaining complex behaviors of large-scale systems (Murray, 2007). As discussed in Reeves et al. (2021), the Barrier3D base model formulation assumes invariant beach width and slope. In reality, beach characteristics can vary over timescales of hours to decades (e.g., O'Dea et al., 2019) and length scales of 10²-10³ m (Vos et al., 2020) in ways that can affect dune growth (Durán & Moore, 2013), dune erosion (Beuzen et al., 2019), and overwash (Donnelly et al., 2006). The base model also does not account for tidal inlet processes, which can constitute a significant proportion of overall transgressive flux in barrier systems (e.g., Leatherman, 1979; Nienhuis & Lorenzo-Trueba, 2019), or other sediment pathways that can lead to deflation of the barrier (Passeri et al., 2020), such as gradients in alongshore transport, breaching, or storm-driven seaward transport. Below, we focus on the most consequential limitations related specifically to the new dynamics introduced by barrier-shrub feedbacks.

First, the model does not simulate a freshwater lens, which changes in size in response to barrier morphologic evolution. Freshwater availability is especially limited in narrow barriers (e.g., Bolyard et al., 1979; Rastet-ter, 1991; Shao et al., 1995). Thus, our parameterization of a shrub elevation range from Hog Island (0.74–1.84 m MHW) may be less suitable for barriers of different morphologies. Incorporation of groundwater dynamics would increase shrub expansion when barriers are wider (and vice versa). Modeled shrubs cannot occur on the beach (because Barrier3D does not model a beach) or, for model simplicity, on the foredune as a result of landward translation of the ocean shoreline. When the shoreline erodes landward one cell width, any shrubs occupying



the front (ocean-side) row of the interior domain are removed from the system entirely. In reality, shrubs can be found on beaches of rapidly transgressing barriers (Figure 1d), and persist until the ocean shoreline recedes far enough landward such that shrubs are impacted by swash zone dynamics. Because this would effectively extend the duration of shrub presence within the barrier system, shrub impacts are likely underestimated in the model in this regard. Shrubs also have the potential to influence aeolian reworking of the barrier interior by causing sand to deposit in and around shrub thickets. Aeolian dynamics of the barrier interior, however, are not included in Barrier3D. In reality, interactions between shrubs and wind-driven sediment transport during interstorm periods may increase surface roughness of the barrier interior, and potentially influence overwash routing.

In the absence of empirical measurements for the effects of shrubs on overwash flow across a barrier, we parameterize Λ , the maximum flow reduction coefficient, with a conservative value that produces qualitatively realistic morphology. In situ measurements of shrub-overwash flow hydrodynamics are needed to increase confidence in the quantitative accuracy of the model results, though sensitivity tests across a broad range of Λ values suggest that the parameter ultimately has limited impact on barrier morphology (Figure S1 in Supporting Information S1). In addition, mechanical damage to shrubs from waves or overwash bores is absent from the model formulation, as is the decay of dead shrubs over time, both of which are difficult processes to parameterize within our exploratory framework. In the absence of these dynamics, shrub longevity and their ability to limit overwash are likely to be marginally overestimated within the model.

We do not include biotic interactions between different plant species, such as facilitation or competition. Similarly, we do not explicitly simulate other finer scale environmental filters of shrub establishment, such as the distance to groundwater, soil salinity, or nutrients, but the collective effects of many of these factors are represented by our larger-scale parameterizations of a threshold dune height for shrub growth and a shrub elevation range. We also do not incorporate positive feedbacks related to shrubs modifying their own microclimate. Shrub thickets modify their microclimate by increasing overnight temperatures as well as soil nutrients and moisture (D'Odorico et al., 2012; Huang et al., 2018; Wood et al., 2020), thereby enhancing shrub growth and facilitating further localized shrub establishment. Incorporating these positive feedbacks within the model would likely increase shrub expansion rates. While our simulations are calibrated for M. cerifera, a common shrub along the Atlantic and Gulf coasts of North America and the dominant species in the VCR, parameterizing the model to represent other shrub species would result in differences to our results but would unlikely change the fundamental conclusions we draw from them. Although our model does not include the possibility for eventual succession of shrubland to maritime forest (the typical climax community on Atlantic barriers), frequent reworking of the barrier and rapid expansion of full monospecific shrub thickets can delay or prevent this successional development (Bissett et al., 2016). Therefore, given projected climate-related increases in atmospheric temperatures, RSLR rates, and storminess, development of new maritime forest is likely to be rare.

5.2. Ecological Impacts on Barrier Morphology

Our results suggest major morphological and behavioral differences between barriers with and without shrubs, as proposed by Zinnert et al. (2017). In our simulations, the time-averaged morphology of barriers with shrubs is narrower and less voluminous than the time-averaged morphology of barriers without shrubs. Across the broad ranges of RSLR and characteristic dune growth rates we simulate, we find that the presence of shrubs reduces barrier widths and volumes by as much as 40%–65%, and that volumes tend to be more distributed toward the front of the barrier. These morphological differences are the result of shrubs limiting overwash flow into and across the barrier in the model: Sediment carried into the barrier interior is deposited over shorter distances from the effects of frictional drag of the shrub thicket on the overwashing flow, which in turn builds topographic relief that limits continued transgressive flux into the barrier interior. The distance over which overwash sediment is deposited is a function of both storm intensity and the density/extent of woody cover.

We also find that barriers with shrubs tend to migrate landward more slowly and are more likely to experience discontinuous retreat. Migration rates in our simulations tend to be approximately 20%–40% slower with shrubs because overwash fluxes are lesser (i.e., less sediment is removed from the shoreface and beach to be deposited as washover). Meanwhile, discontinuous retreat is more likely because shrubs extend the duration of immobile periods by delaying the onset of overwash flux after dunes are lost, providing prolonged periods of immobility needed to produce discontinuous behavior. This lag lasts until shrub cover has sufficiently decreased, and can extend periods of immobility in our simulations by as much as 340%. However, under conditions that tend to result in



relatively immobile barriers (i.e., fast dune growth rates, low RSLR rates), this lag effect can potentially prevent any transgressive periods from occurring at all, which tends to reduce the likelihood of discontinuous behavior in favor of continuous immobility and increase the likelihood of barrier drowning. At high dune growth rates in our simulations, barriers with shrubs are twice as vulnerable to drowning within 1,000 years as barriers without.

Our results suggest that interior ecology is a critical component of barrier evolution on timescales of a single storm to many centuries, complementing recent studies by Zinnert et al. (2016, 2017, 2019). In our simulations, dune growth rates and the presence or absence of shrubs are the primary drivers of barrier width and vulnerability to drowning, not RSLR. This emphasizes the prominent role of internal dynamics – as opposed to external forcing – in controlling barrier evolution (Ciarletta et al., 2019; Lorenzo-Trueba & Ashton, 2014; Reeves et al., 2021). In this way, the impacts of shrubs are similar to those of dunes (Reeves et al., 2021) and human infrastructure (Lazarus et al., 2021; Rogers et al., 2015), two additional suites of processes internal to barrier systems that can restrict sediment delivery into and across the barrier interior.

The significant morphologic and behavioral differences we observe in our simulations between barriers with shrubs and without suggest major disparities in barrier functionality and services, with potentially broad impacts on the ecological, physical, and human components of the coastal barrier system. For example, barriers with shrubs may be unable to support fringing back-barrier marshes that rely on overwash sediment deposition to keep pace with RSLR (Walters et al., 2014; Walters & Kirwan, 2016); barriers that are narrow in the presence of shrubs may have limited freshwater lenses (Bolyard et al., 1979), and might support less rich ecosystems and human activities; and, the tendency for nonlinear behavior and drowning of barriers with shrubs could lead to cascading state changes across the entire coastal system (e.g., Deaton et al., 2017; FitzGerald et al., 2018). With shrub expansion worldwide linked to global warming, these internal shrub dynamics may become increasingly important in the future across barrier systems with climatologies that had precluded shrub establishment in the past. However, as sea level rises and rates of dune growth (relative to the frequency of intense storms) decrease, our results suggest that the impacts of shrubs on barrier morphology and behavior (relative to barriers without shrubs) may become less influential because of a consequent decline in shrub cover.

Our results also highlight complex, sometimes counterintuitive feedbacks and relationships between shrub ecology and barrier geomorphology. For instance, the tendency for higher rates of RSLR to reduce barrier widths is reversed when shrubs are included in the simulations. In addition, despite the fact that sufficiently tall dunes are required for shrub growth, faster dune growth rates typically lead to less shrub cover over long time scales because tall dunes limit overwash deposition needed to offset habitat loss from passive drowning of the back-barrier shoreline via RSLR. Further, the average duration of immobile periods tends to decrease with higher dune growth rates in the presence of shrubs. These complexities emphasize the importance of explicitly modeling barrier ecology and its interactions with the physical environment when exploring long-term barrier evolution, as opposed to using static representations of land-cover that would be unable to capture these dynamics.

5.3. Geomorphological Impacts on Shrub Expansion

The substantial variance in shrub expansion rates in our simulations can be attributed to the stochasticity of not only shrub seed dispersal but also dune-overwash dynamics. Our results suggest that dune and overwash dynamics are important for understanding or predicting the rate and pattern of shrub expansion. Over annual to decadal timescales, we find that storm-related disturbances of dune erosion and overwash are associated with slower expansion rates and more years without any positive translation alongshore of the shrub front. Higher rates of RSLR and slower dune growth typically induce more frequent disturbance events. In this way, storms act as a destructive force that both kills and/or erodes away established shrubs, which has the additional effect of reducing the number of seeds dispersed, and prevents the establishment of shrub seedlings by enabling unsuitable environmental conditions. Rapid, uninterrupted shrub expansion is therefore most likely to occur across barriers that are disturbance resistant, such as Hog Island (Wolner et al., 2013) where shrubs have expanded at surprisingly rapid rates of up to 300 m yr⁻¹ (Woods et al., 2019).

However, over longer, decadal to centurial time scales, the physical impacts of storms are both destructive and constructive in the context of shrub ecology. Washover deposition can both maintain existing shrub habitat by raising interior elevations relative to sea level, and build new shrub habitat by extending the back-barrier shore-line landward. These constructive tendencies in the model tend to contribute (over long timescales) toward habitat



exchange, a component of habitat change in which the loss of a habitat type is offset by gain of that habitat elsewhere; Enwright et al. (2021) find a potential relationship between barrier migration and barrier habitat exchange from change component analysis of Dauphin Island, Alabama, USA. Our simulations suggest that, over multiple cycles of shrub expansion and mortality, shrub cover (and, by extension, shrub impacts) is often greatest when the constructive and destructive tendencies of overwash processes are balanced. Although warming macroclimate temperatures are likely to stimulate shrub expansion in the future, our results suggest this tendency may be tempered in part by increases in disturbances related to higher RSLR rates and slower dune growth rates (i.e., relative to increases in the frequency of more intense storm events) – or the greater vulnerability of barriers with shrubs to drowning. While we examined shrub expansion across a range of RSLR and characteristic dune growth rates, future work should include the effects of climate warming on ecological aspects of shrub expansion (e.g., seed production, germination rate) in addition to these physical drivers.

5.4. Comparisons to Observations

Comparing our model results to observations is important for understanding the explanatory power of the model. Making such comparisons, however, is challenging because the durations of our simulations and the characteristic timescales of dynamics we observe within them are often much greater than the durations of most observational time series. Nevertheless, although quantitative comparisons are inappropriate given the exploratory nature of our model, key aspects of our results are qualitatively consistent with documented real-world behavior.

Comparisons between shrub expansion in our model and on Hog Island in the VCR suggest that our new shrub module is capable of producing realistic shrub behavior. Shrub expansion rates observed on Hog Island from 1984 to 2010 average 164 m yr⁻¹ and reach a maximum of 300 m yr⁻¹ (Woods et al., 2019), well within the range of average expansion rates from our simulations of 0-334 m yr⁻¹ described in Section 4 above. Additionally, Woods et al. (2019) observe a logarithmic trend in shrub front expansion over time on Hog Island, a dynamic that is captured in our model simulations as well (Figure 8). Our model results also identify conditions in which barrier processes construct new shrub habitat; this is supported by the empirical observations of Young et al. (2007), who find a positive relationship between change in subaerial barrier area and change in shrub cover for the larger barriers of the VCR, and shrub expansion onto newly-created washover deposits has been observed on Fire Island, NY (Leatherman, 1985), Nauset Spit, MA (Zaremba & Leatherman, 1986), and Smith Island, VA (Young et al., 2007).

In our model simulations, barriers with shrubs tend to be narrower, less voluminous, and be more strongly skewed volumetrically toward the ocean. Woody vegetation has been observed to block overwash flow and build topographic relief both in natural barrier settings (Claudino-Sales et al., 2008; Wang & Horwitz, 2007) and within physical models (Kobayashi et al., 2013). In an analysis of remotely-sensed observations from VCR barrier islands, Zinnert et al. (2019) find an important relationship between woody cover and back-barrier shoreline extension, suggesting that shrub cover reduces the landward translation of back-barrier shorelines by decreasing overwash sediment delivery across the barrier, thereby leading to barrier narrowing.

Our model results also suggest that barriers with shrubs are more likely to experience discontinuous retreat and longer periods of immobility because shrubs delay the onset of transgression after dunes transition to a low-elevation state. This lag effect is supported by observational evidence from Cobb Island in the VCR of a temporal correlation between the onset of rapid landward translation of the back-barrier shoreline and the loss of woody cover from erosion (Zinnert et al., 2019). From 1984 to 1998, shrubs expanded rapidly across Cobb Island during a prolonged period of stability. From 1998 to 2011, extensive shoreline erosion occurred (likely because of a change in sediment supply or dune elevation state) but the back-barrier shoreline experienced little change, leading to the barrier losing 63% of its subaerial area (which can be taken as a proxy for barrier width and volume). Back-barrier shoreline extension increased rapidly between 2011 and 2016 once shrub cover was reduced to nearly zero by erosion and exposure to the open ocean.

The connections we have highlighted in this section between model findings and empirical observations suggest our model qualitatively captures real-world barrier-shrub behavior. Observational analyses beyond the scope of this study are needed to continue testing the ideas presented herein. More broadly, our results suggest the need for more observational research that integrates interior barrier dynamics with geomorphological change over time.



6. Conclusions

We introduce shrub expansion and mortality processes within an exploratory model of barrier morphodynamics, and point to the essential role of interior ecology in barrier evolution. Our model simulations suggest that barriers with shrubs, relative to those without, tend to be narrower and migrate landward more slowly, with subaerial volume distributed more toward the front (ocean-side) of the barrier. In addition, in our model simulations, shrubs generally increase the likelihood of discontinuous retreat by providing prolonged cessations in overwash flux needed to produce periods of immobility. However, under conditions where periods of immobility are common, shrubs can altogether inhibit periods of transgression that are needed for barriers to maintain elevation relative to sea level, leading to increased vulnerability to drowning in the presence of shrubs. Additionally, our model experiments suggest that shrub expansion is highly dependent on geomorphological barrier processes. Slower shrub front expansion rates are correlated with greater dune erosion and overwash disturbance, which occur more often, on average, at higher rates of RSLR and slower rates of dune growth. In our model, shrub-barrier couplings lead to complex and sometimes counterintuitive outcomes that emphasize the need to explicitly account for ecological processes and their interactions with the physical environment when exploring decadal to centurial barrier evolution.

Data Availability Statement

Barrier3D (Reeves, 2021) is available for download from the online Community Surface Dynamics Modeling System model repository at https://csdms.colorado.edu/wiki/Model:Barrier3D. Data for experimental results are in repository at https://doi.org/10.6073/pasta/5b59143f40b0b73f01337281f1ae1b51.

References

- Battaglia, L. L., Denslow, J. S., & Hargis, T. G. (2007). Does woody species establishment alter herbaceous community composition of freshwater floating marshes? *Journal of Coastal Research*, 23(6), 1580–1587. https://doi.org/10.2112/06-0766.1
- Beuzen, T., Harley, M. D., Splinter, K. D., & Turner, I. L. (2019). Controls of variability in berm and dune storm erosion. Journal of Geophysical Research: Earth Surface, 124(11), 2647–2665. https://doi.org/10.1029/2019JF005184
- Biel, R. G., Hacker, S. D., & Ruggiero, P. (2019). Elucidating coastal foredune ecomorphodynamics in the US Pacific northwest via Bayesian networks. Journal of Geophysical Research: Earth Surface, 124(7), 1919–1938. https://doi.org/10.1029/2018JF004758
- Bissett, S. N., Zinnert, J. C., & Young, D. R. (2016). Woody expansion facilitates liana expansion and affects physical structure in temperate coastal communities. *Ecosphere*, 7(6), e01383. https://doi.org/10.1002/ecs2.1383
- Bolyard, T. H., Hornberger, G. M., Dolan, R., & Hayden, B. P. (1979). Freshwater reserves of mid-Atlantic coast barrier islands. *Environmental Geology*, 3(1), 1–11. https://doi.org/10.1007/BF02423273
- Bruun, P. (1962). Sea-level rise as a cause of shore erosion. Journal of the Waterways and Harbors Division, 88, 117–130. https://doi.org/10.1061/jwheau.0000252
- Carr, J., Mariotti, G., Fahgerazzi, S., McGlathery, K., & Wiberg, P. (2018). Exploring the impacts of seagrass on coupled marsh-tidal flat morphodynamics. Frontiers in Environmental Science, 6(92). https://doi.org/10.3389/fenvs.2018.00092
- Chow, V. T. (1959). Open channel hydraulics. Civil engineering series. McGraw Hill.

Ciarletta, D. J., Lorenzo-Trueba, J., & Ashton, A. D. (2019). Mechanism for retreating barriers to autogenically form periodic deposits on continental shelves. *Geology*, 47(3), 239–242. https://doi.org/10.1130/g45519.1

- Claudino-Sales, V., Wang, P., & Horwitz, M. H. (2008). Factors controlling the survival of coastal dunes during multiple hurricane impacts in 2004 and 2005: Santa Rosa barrier island, Florida. *Geomorphology*, 95(3–4), 295–315. https://doi.org/10.1016/j.geomorph.2007.06.004
- Conn, C. E., & Day, F. P. (1993). Belowground biomass patterns on a coastal barrier island in Virginia. Bulletin of the Torrey Botanical Club, 120(2), 121–127. https://doi.org/10.2307/2996941
- Crawford, E. R., & Yaoung, D. R. (1998). Comparison of gaps and intact shrub thickets on an Atlantic coast barrier island. *The American Midland Naturalist*, 140(1), 68–77. https://doi.org/10.1674/0003-0031(1998)140
- Deaton, C. D., Hein, C. J., & Kirwan, M. L. (2017). Barrier island migration dominates ecogeomorphic feedbacks and drives salt marsh loss along the Virginia Atlantic Coast, USA. *Geology*, 45(2), 123–126. https://doi.org/10.1130/G38459.1
- D'Odorico, P., Okin, G. S., & Bestelmeyer, B. T. (2012). A synthetic review of feedbacks and drivers of shrub encroachment in arid grasslands. *Ecohydrology*, 5(5), 520–530. https://doi.org/10.1002/eco.259
- Dolan, R., & Godfrey, P. (1973). Effects of hurricane ginger on the barrier islands of North Carolina. *Geological Society of America Bulletin*, 84(4), 1329–1334. https://doi.org/10.1130/0016-7606(1973)84<1329
- Donnelly, C., Kraus, N., & Larson, M. (2006). State of knowledge on measurement and modeling of coastal overwash. Journal of Coastal Research, 22(4), 965–991. https://doi.org/10.2112/04-0431.1
- Durán, O., & Moore, L. J. (2013). Vegetation controls on the maximum size of coastal dunes. Proceedings of the national academy of Sciences of the United States of America, 110(43), 17217–17222. https://doi.org/10.1073/pnas.1307580110
- Durán Vinent, O., & Moore, L. J. (2015). Barrier island bistability induced by biophysical interactions. *Nature Climate Change*, 5(2), 1758–6798. https://doi.org/10.1038/nclimate2474
- Ehrenfeld, J. G. (1990). Dynamics and processes of barrier-island vegetation. Aquatic Sciences, 2(3-4), 437-480.
- Enwright, N. M., Wang, L., Borchert, S. M., Day, R. H., Feher, L. C., & Osland, M. J. (2019). Advancing barrier island habitat mapping using landscape position information. Progress in Physical Geography: Earth and Environment, 43(3), 425–450. https://doi.org/10.1177/0309133319839922

Acknowledgments

We thank the Associate Editor and three reviewers for their valuable feedback on this manuscript. This work is supported by the Virginia Coast Reserve Long-Term Ecological Research Program (National Science Foundation DEB-1832221) via a sub-award to L. J. Moore and J. C. Zinnert and the Preston Jones and Mary Elizabeth Frances Dean Martin Fellowship Fund from the Department of Earth, Marine and Environmental Sciences at the University of North Carolina at Chapel Hill. The authors acknowledge computing time on the CU-CSDMS High-Performance Computing Cluster.



- Enwright, N. M., Wang, L., Dalyander, P. S., Wang, H., Osland, M. J., Mickey, R. C., et al. (2021). Assessing habitat change and migration of barrier islands. *Estuaries and Coasts*, 44(8), 2073–2086. https://doi.org/10.1007/s12237-021-00971-w
- FitzGerald, D. M., Hein, C. J., Hughes, Z., Kulp, M., Georgiou, I., & Miner, M. (2018). Runaway barrier island transgression concept: Global case studies. In L. J. Moore & A. B. Murray (Eds.), *Barrier dynamics and response to changing climate* (pp. 3–56). Springer International Publishing. https://doi.org/10.1007/978-3-319-68086-6_1
- Godfrey, P. J. (1977). Climate, plant response and development of dunes on barrier beaches along the U.S. East Coast. International Journal of Biometeorology, 21(3), 203–215. https://doi.org/10.1007/bf01552874
- Godfrey, P. J., Leatherman, S. P., & Zaremba, R. (1979). A geobotanical approach to classification of barrier beach systems. In S. P. Leatherman (Ed.), Barrier islands from the Gulf of St. Lawrence to the Gulf of Mexico (pp. 99–126). Academic Press.
- Goldstein, E. B., & Moore, L. J. (2016). Stability and bistability in a one-dimensional model of coastal foredune height. Journal of Geophysical Research: Earth Surface, 121(5), 964–977. https://doi.org/10.1002/2015JF003783
- Goldstein, E. B., Moore, L. J., & Durán Vinent, O. (2017). Lateral vegetation growth rates exert control on coastal foredune hummockiness and coalescing time. *Earth Surface Dynamics*, 5(3), 417–427. https://doi.org/10.5194/esurf-5-417-2017
- Goldstein, E. B., Mullins, E. V., Moore, L. J., Biel, R. G., Brown, J. K., Hacker, S. D., et al. (2018). Literature-based latitudinal distribution and possible range shifts of two US East Coast dune grass species (Uniola paniculata and Ammophila breviligulata). *PeerJ*, 6, e4932. https://doi. org/10.7717/peerj.4932
- Hacker, S. D., Jay, K. R., Cohn, N., Goldstein, E. B., Hovenga, P. A., Itzkin, M., et al. (2019). Species-specific functional morphology of four US Atlantic coast dune grasses: Biogeographic implications for dune shape and coastal protection. *Diversity*, 11(5), 82. https://doi.org/10.3390/ d11050082
- Herrmann, J. D., Carlo, T. A., Brudvig, L. A., Damschen, E. I., Haddad, N. M., Levey, D. J., et al. (2016). Connectivity from a different perspective: Comparing seed dispersal kernels in connected vs. Unfragmented landscapes. *Ecology*, 97(5), 1274–1282. https://doi.org/10.1890/15-0734.1
- Hesp, P. A. (2002). Foredunes and blowouts: Initiation, geomorphology, and dynamics. *Geomorphology*, 48(1–3), 245–268. https://doi.org/10.1016/S0169-555X(02)00184-8
- Hokkanen, M. (2013). Environmental influences on sex ratio and spatial distribution of dioecious Morella cerifera L. on a Virginia barrier island (Master's thesis). Virginia Commonwealth University. Retrieved from https://doi.org/10.25772/5HF6-NB36
- Houser, C., Barrineau, P., Hammond, B., Saari, B., Rentschler, E., Trimble, S., et al. (2018). Role of the foredune in controlling barrier island response to sea level rise. In L. J. Moore & A. B. Murray (Eds.), *Barrier dynamics and response to changing climate* (pp. 175–207). Springer International Publishing. https://doi.org/10.1007/978-3-319-68086-6_6
- Houser, C., Wernette, P., Rentschlar, E., Jones, H., Hammond, B., & Trimble, S. (2015). Post-storm beach and dune recovery: Implications for barrier island resilience. *Geomorphology*, 234(1), 54–63. https://doi.org/10.1016/j.geomorph.2014.12.044
- Huang, H., Anderegg, L. D., Dawson, T. E., Mote, S., & D'Odorico, P. (2020). Critical transition to woody plant dominance through microclimate feedbacks in North American coastal ecosystems. *Ecology*, 101(9), e03107. https://doi.org/10.1002/ecy.3107
- Huang, H., Zinnert, J. C., Wood, L. K., Young, D. R., & D'Odorico, P. (2018). Non-linear shift from grassland to shrubland in temperate barrier islands. *Ecology*, 99(7), 1671–1681. https://doi.org/10.1002/ecy.2383
- Keijsers, J. G. S., Groot, A. V. D., & Riksen, M. J. P. M. (2016). Modeling the biogeomorphic evolution of coastal dunes in response to climate change. *Journal of Geophysical Research: Earth Surface*, 121(6), 1161–1181. https://doi.org/10.1002/2015JF003815
- Keller, D. N. (2020). Stimulated growth response to sand burial of a coastal shrub (Master's thesis). Virginia Commonwealth University. Retrieved from https://scholarscompass.vcu.edu/etd/6451
- Kirwan, M. L., & Murray, A. B. (2007). A coupled geomorphic and ecological model of tidal marsh evolution. Proceedings of the National Academy of Sciences, 104(15), 6118–6122. https://doi.org/10.1073/pnas.0700958104
- Knutson, T., Camargo, S. J., Chan, J. C., Emanuel, K., Ho, C. H., Kossin, J., et al. (2020). Tropical cyclones and climate change assessment: Part II: Projected response to anthropogenic warming. *Bulletin of the American Meteorological Society*, 101(3), E303–E322. https://doi. org/10.1175/BAMS-D-18-0194.1
- Kobayashi, N., Gralher, C., & Do, K. (2013). Effects of woody plants on dune erosion and overwash. Journal of Waterway, Port, Coastal, and Ocean Engineering, 139(6), 466–472. https://doi.org/10.1061/(ASCE)WW.1943-5460.0000200
- Kwit, C., Levey, D. J., Greenberg, C. H., Pearson, S. F., McCarty, J. P., & Sargent, S. (2004). Cold temperature increases winter fruit removal rate of a bird-dispersed shrub. *Oecologia*, 139(1), 30–34. https://doi.org/10.1007/s00442-003-1470-6
- Lazarus, E. D., Goldstein, E. B., Taylor, L. A., & Williams, H. E. (2021). Comparing patterns of hurricane washover into built and unbuilt environments. *Earth's Future*, 9(3), e2020EF001818. https://doi.org/10.1029/2020EF001818
- Leatherman, S. P. (1979). Migration of Assateague Island, Maryland, by inlet and overwash processes. *Geology*, 7, 104–107. https://doi.org/10.1130/0091-7613(1979)7<104
- Leatherman, S. P. (1985). Geomorphic and stratigraphic analysis of fire island, New York. *Marine Geology*, 63(1-4), 173–195. https://doi. org/10.1016/0025-3227(85)90083-0
- Long, J. W., de Bakker, A. T., & Plant, N. G. (2014). Scaling coastal dune elevation changes across storm-impact regimes. *Geophysical Research Letters*, 41(8), 2899–2906. https://doi.org/10.1002/2014GL059616
- Lorenzo-Trueba, J., & Ashton, A. (2014). Rollover, drowning, and discontinuous retreat: Distinct modes of barrier response to sea-level rise arising from a simple morphodynamic model. *Journal of Geophysical Research: Earth Surface*, 119(4), 779–801. https://doi.org/10.1002/2013JF002941
- Lucas, K. L., & Carter, G. A. (2010). Decadal changes in habitat-type coverage on Horn Island, Mississippi, USA. Journal of Coastal Research, 26(6), 1142–1148. https://doi.org/10.2112/JCOASTRES-D-09-00018.1
- Magliocca, N. R., McNamara, D. E., & Murray, A. B. (2011). Long-term, large-scale morphodynamic effects of artificial dune construction along a barrier island coastline. *Journal of Coastal Research*, 27(5), 918–930. https://doi.org/10.2112/JCOASTRES-D-10-00088.1
- Mellett, C. L., & Plater, A. J. (2018). Drowned barriers as archives of coastal-response to sea-level rise. In L. J. Moore & A. B. Murray (Eds.), Barrier dynamics and response to changing climate (pp. 57–89). Springer International Publishing. https://doi.org/10.1007/978-3-319-68086-6_2
 - Miller, D. L., Thetford, M., & Schneider, M. (2008). Distance from the Gulf influences survival and growth of three barrier island dune plants. *Journal of Coastal Research*, 4(24), 261–266. https://doi.org/10.2112/07-0914.1
 - Moore, L. J., List, J. H., Williams, S. J., & Stolper, D. (2010). Complexities in barrier island response to sea level rise: Insights from numerical model experiments, North Carolina outer banks. *Journal of Geophysical Research*, 115(F3), F03004. https://doi.org/10.1029/2009JF001299
 - Morris, J. T., Sundareshwar, P. V., Nietch, C. T., Kjerfve, B., & Cahoon, D. R. (2002). Responses of coastal wetlands to rising sea level. *Ecology*, 83(10), 2869–2877. https://doi.org/10.1890/0012-9658(2002)083%5B2869
 - Morton, R. A., & Sallenger, A. H., Jr. (2003). Morphological impacts of extreme storms on sandy beaches and barriers. *Journal of Coastal Research*, 560–573.



- Murray, A. B. (2003). Contrasting the goals, strategies, and predictions associated with simplified numerical models and detailed simulations. In Prediction in geomorphology (pp. 151–165). AGU.
- Murray, A. B. (2007). Reducing model complexity for explanation and prediction. Geomorphology, 90(3–4), 178–191. https://doi.org/10.1016/j.geomorph.2006.10.020
- Murray, A. B. (2013). Which models are good (enough), and when? In J. F. Shroder (Ed.), *Treatise on geomorphology* (pp. 50–58). Elsevier—Academic Press.
- Murray, A. B., & Paola, C. (1994). A cellular model of braided rivers. Nature, 371(6492), 54-57. https://doi.org/10.1038/371054a0
- Murray, A. B., & Paola, C. (1997). Properties of a cellular braided-stream model. Earth Surface Processes and Landforms, 22(11), 1001–1025. https://doi.org/10.1002/(SICI)1096-9837
- Nienhuis, J. H., & Lorenzo-Trueba, J. (2019). Can barrier islands survive sea-level rise? Quantifying the relative role of tidal inlets and overwash deposition. *Geophysical Research Letters*, 46(24), 14613–14621. https://doi.org/10.1029/2019GL085524
- NOAA. (2017). USACE NCMP topobathy lidar DEM: East coast. NOAA National Centers for Environmental Information. Retrieved from https://www.fisheries.noaa.gov/inport/item/52446
- NOAA. (2021). Sea level trends: Wachapreague, Virginia 8631044. NOAA Tides and Currents. Retrieved from https://tidesandcurrents.noaa.gov/sltrends/
- O'Dea, A., Brodie, K. L., & Hartzell, P. (2019). Continuous coastal monitoring with an automated terrestrial lidar scanner. *Journal of Marine Science and Engineering*, 7(2), 37. https://doi.org/10.3390/jmse7020037
- Orth, R. J., & McGlathery, K. J. (2012). Eelgrass recovery in the coastal bays of the Virginia Coast Reserve, USA. Marine Ecology Progress Series, 448, 173–176. https://doi.org/10.3354/meps09596
- Osland, M. J., Enwright, N. M., Day, R. H., Gabler, C. A., Stagg, C. L., & Grace, J. B. (2016). Beyond just sea-level rise: Considering macroclimatic drivers within coastal wetland vulnerability assessments to climate change. *Global Change Biology*, 22(1), 1–11. https://doi.org/10.1111/ gcb.13084
- Oster, D., & Moore, L. (2009). Beach Morphology of the Virginia Barrier Islands 1998, 2005 and 2009. Virginia Coast Reserve Long-Term Ecological Research Project Data Publication knb-lter-vcr.279.1. https://doi.org/10.6073/pasta/95bf349362e134e6730b9568f1946b1d
- Passeri, D. L., Dalyander, P. S., Long, J. W., Mickey, R. C., Jenkins, R. L., III, Thompson, D. M., et al. (2020). The roles of storminess and sea level rise in decadal barrier island evolution. *Geophysical Research Letters*, 47(18), e2020GL089370. https://doi.org/10.1029/2020GL089370
- Passeri, D. L., Long, J. W., Plant, N. G., Bilskie, M. V., & Hagen, S. C. (2018). The influence of bed friction variability due to land cover on storm-driven barrier island morphodynamics. *Coastal Engineering*, 132, 82–94. https://doi.org/10.1016/j.coastaleng.2017.11.005
- Peters, D. P., Bestelmeyer, B. T., Herrick, J. E., Fredrickson, E. L., Monger, H. C., & Havstad, K. M. (2006). Disentangling complex landscapes: New insights into arid and semiarid system dynamics. *BioScience*, 56(6), 491–501. https://doi.org/10.1641/0006-3568(2006)56[491:DCLNII]2.0.CO;2
- Rastetter, E. B. (1991). A spatially explicit model of vegetation-habitat interactions on barrier islands. In M. G. Turner & R. H. Gardiner (Eds.), Quantitative methods in landscape ecology. Ecological studies (Vol. 82, pp. 353–378). Springer-Verlag.
- Reeves, I. (2021). Barrier3D v1.0. https://doi.org/10.5281/zenodo.4730989
- Reeves, I. R. B., Moore, L. J., Goldstein, E. B., Murray, A. B., Carr, J. A., & Kirwan, M. L. (2020). Impacts of seagrass dynamics on the coupled long-term evolution of barrier- marsh-bay systems. *Journal of Geophysical Research: Biogeosciences*, 125(2), e2019JG005416. https://doi. org/10.1029/2019JG005416
- Reeves, I. R. B., Moore, L. J., Murray, A. B., Anarde, K. A., & Goldstein, E. B. (2021). Dune dynamics drive discontinuous barrier retreat. Geophysical Research Letters, 48(13), e2021GL092958. https://doi.org/10.1029/2021GL092958
- Rogers, L. J., Moore, L. J., Goldstein, E. B., Hein, C. J., Lorenzo-Trueba, J., & Ashton, A. D. (2015). Anthropogenic controls on overwash deposition: Evidence and consequences. *Journal of Geophysical Research: Earth Surface*, 120(12), 2609–2624. https://doi.org/10.1002/2015JF003634
- Roman, C. T., & Nordstrom, K. F. (1988). The effect of erosion rate on vegetation patterns of an East Coast Barrier Island. *Estuarine, Coastal and Shelf Science*, 26(3), 233–242. https://doi.org/10.1016/0272-7714(88)90062-5
- Sallenger, A. H. (2000). Storm impact scale for barrier islands. Journal of Coastal Research, 16, 890-895.
- Schwartz, R. K. (1982). Bedform and stratification characteristics of some modern small-scale washover sand bodies. Sedimentology, 29(6), 835–849. https://doi.org/10.1111/j.1365-3091.1982.tb00087.x
- Shao, G., Shugart, H. H., & Young, D. R. (1995). Simulation of transpiration sensitivity to environmental changes for shrub (Myrica cerifera) thickets on a Virginia barrier island. *Ecological Modelling*, 78(3), 235–248. https://doi.org/10.1016/0304-3800(93)E0091-G
- Shao, G., Young, D. R., Porter, J. H., & Hayden, B. P. (1998). An integration of remote sensing and GIS to examine the responses of shrub thicket distributions to shoreline changes on Virginia barrier islands. *Journal of Coastal Research*, 14, 299–307.
- Shaver, G. R., Bret-Harte, M. S., Jones, M. H., Johnstone, J., Gough, L., Laundre, J., & Chapin, F. S., III. (2001). Species composition interacts with fertilizer to control long-term change in tundra productivity. *Ecology*, 82(11), 3163–3181. https://doi.org/10.1890/0012-9658(2001)082 [3163:SCIWFT]2.0.CO;2
- Shiflett, S. A., & Young, D. R. (2010). Avian seed dispersal on Virginia barrier islands: Potential influence on vegetation community structure and patch dynamics. *The American Midland Naturalist*, 164(1), 91–106. https://doi.org/10.1674/0003-0031-164.1.91
- Stallins, J. A., & Parker, A. J. (2003). The influence of complex systems interactions on barrier island dune vegetation pattern and process. *Annals of the Association of American Geographers*, 93(1), 13–29. https://doi.org/10.1111/1467-8306.93102
- Stevens, N., Lehmann, C. E., Murphy, B. P., & Durigan, G. (2017). Savanna woody encroachment is widespread across three continents. *Global Change Biology*, 23(1), 235–244. https://doi.org/10.1111/gcb.13409
- Stutz, M. L., & Pilkey, O. H. (2011). Open-ocean barrier islands: Global influence of climatic, oceanographic, and depositional settings. Journal of Coastal Research, 27(2), 207–222. https://doi.org/10.2112/09-1190.1
- Sweet, W. V., Kopp, R. E., Weaver, C. P., Obeysekera, J., Horton, R., Thieler, E. R., & Zervas, C. (2017). Global and regional sea level rise scenarios for the United States (Technical Report NOS CO-OPS 083). Silver Spring. National oceanic and Atmospheric Administration.
- Tape, K., Sturm, M., & Racine, C. (2006). The evidence for shrub expansion in northern Alaska and the Pan-Arctic. Global Change Biology, 12(4), 686–702. https://doi.org/10.1111/j.1365-2486.2006.01128.x
 - Tolliver, K. S., Martin, D. W., & Young, D. R. (1997). Freshwater and saltwater flooding response for woody species common to barrier island swales. Wetlands, 17(1), 10–18. https://doi.org/10.1007/BF03160714
 - Van Auken, O. W. (2000). Shrub invasions of North American semiarid grasslands. Annual Review of Ecology and Systematics, 31(1), 197–215. https://doi.org/10.1146/annurey.ecolsys.31.1.197
 - van IJzendoorn, C. O., de Vries, S., Hallin, C., & Hesp, P. A. (2021). Sea level rise outpaced by vertical dune toe translation on prograding coasts. *Scientific Reports*, 11(1), 12792. https://doi.org/10.1038/s41598-021-92150-x
 - Vos, K., Harley, M. D., Splinter, K. D., Walker, A., & Turner, I. L. (2020). Beach slopes from satellite-derived shorelines. *Geophysical Research Letters*, 47(14), e2020GL088365. https://doi.org/10.1029/2020GL088365



- Wahl, T., Plant, N. G., & Long, J. W. (2016). Probabilistic assessment of erosion and flooding risk in the northern Gulf of Mexico. Journal of Geophysical Research: Oceans, 121(5), 3029–3043. https://doi.org/10.1002/2015JC011482
- Walters, D., Moore, L. J., Durán Vinent, O., Fagherazzi, S., & Mariotti, G. (2014). Interactions between barrier islands and backbarrier marshes affect island system response to sea level rise: Insights from a coupled model. *Journal of Geophysical Research: Earth Surface*, 119(9), 2013–2031. https://doi.org/10.1002/2014JF003091
- Walters, D. C., & Kirwan, M. L. (2016). Optimal hurricane overwash thickness for maximizing marsh resilience to sea level rise. *Ecology and Evolution*, 6(9), 2948–2956. https://doi.org/10.1002/ece3.2024
- Wang, P., & Horwitz, M. H. (2007). Erosional and depositional characteristics of regional overwash deposits caused by multiple hurricanes. Sedimentology, 54(3), 545–564. https://doi.org/10.1111/j.1365-3091.2006.00848.x
- Williams, H. F. L. (2015). Contrasting styles of hurricane irene washover sedimentation on three east coast barrier islands: Cape lookout, North Carolina; Assateague Island, Virginia; and fire island, New York. *Geomorphology*, 231, 182–192. https://doi.org/10.1016/j. geomorph.2014.11.027
- Wolner, C. W. V., Moore, L. J., Young, D. R., Brantley, S. T., Bissett, S. N., & Mcbride, R. A. (2013). Ecomorphodynamic feedbacks and barrier island response to disturbance: Insights from the Virginia Barrier Islands. *Mid-Atlantic Bight, U.S.A. Geomorphology*, 199, 115–128. https:// doi.org/10.1016/j.geomorph.2013.03.035
- Wood, L. K., Hays, S., & Zinnert, J. C. (2020). Decreased temperature variance associated with biotic composition enhances coastal shrub encroachment. *Scientific Reports*, 10(1), 8210. https://doi.org/10.1038/s41598-020-65161-3
- Woods, N. N., Dows, B. L., Goldstein, E. B., Moore, L. J., Young, D. R., & Zinnert, J. C. (2019). Interaction of seed dispersal and environmental filtering affects woody encroachment patterns in coastal grassland. *Ecosphere*, 10(7), e02818. https://doi.org/10.1002/ecs2.2818
- Young, D. R., Brantley, S. T., Zinnert, J. C., & Vick, J. K. (2011). Landscape position and habitat polygons in a dynamic coastal environment. *Ecosphere*, 2(6), 1–15. https://doi.org/10.1890/ES10-00186.1
- Young, D. R., Erickson, D. L., & Semones, S. W. (1994). Salinity and the small-scale distribution of three barrier island shrubs. Canadian Journal of Botany, 72(9), 1365–1372. https://doi.org/10.1139/b94-167
- Young, D. R., Porter, J. H., Bachmann, C. M., Shao, G., Fusina, R. A., Bowles, J. H., et al. (2007). Cross-scale patterns in shrub thicket dynamics in the Virginia barrier complex. *Ecosystems*, 10(5), 854–863. https://doi.org/10.1007/s10021-007-9084-1
- Zaremba, R. E., & Leatherman, S. P. (1986). Vegetative physiographic analysis of a US northern barrier system. *Environmental Geology and Water Sciences*, 8(4), 193–207. https://doi.org/10.1007/bf02524947
- Zinnert, J. C., Shiflett, S. A., Via, S., Bissett, S., Dows, B., Manley, P., & Young, D. R. (2016). Spatial-temporal dynamics in barrier island upland vegetation: The overlooked coastal landscape. *Ecosystems*, 19(4), 685–697. https://doi.org/10.1007/s10021-016-9961-6
- Zinnert, J. C., Stallins, J. A., Brantley, S. T., & Young, D. R. (2017). Crossing scales: The complexity of barrier-island processes for predicting future change. *BioScience*, 67(1), 39–52. https://doi.org/10.1093/biosci/biw154
- Zinnert, J. C., Via, S. M., Nettleton, B. P., Tuley, P. A., Moore, L. J., & Stallins, J. A. (2019). Connectivity in coastal systems: Barrier island vegetation influences upland migration in a changing climate. *Global Change Biology*, 25(7), 2419–2430. https://doi.org/10.1111/gcb.14635